
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Thu Mar 17 17:03:02 GMT 2011
Year: 11  Doy: 076
Observer: stueben
WEATHER COMMENT: Thu Mar 17 17:10:50 GMT 2011
Thin to moderate density cirrus covers most of the sky at present, wind 
southeast 20-25 mph, temp 42F.
___end___
Thu Mar 17 17:23:02 GMT 2011 H-ALPHA Start Patrol
Thu Mar 17 17:41:02 GMT 2011: PSPT Start Patrol
Thu Mar 17 17:58:09 GMT 2011    CHIP     Start Patrol
Thu Mar 17 18:01:53 GMT 2011    CHIP     LSD
**GENERAL PROBLEM COMMENT BY ALLEN**: Thu Mar 17 17:59:07 GMT 2011
The comp and halpa pcs were down this morning when I arrived. This may have been
 caused by an automatic microsoft update which caused a reboot. These pcs don’t
reliably warm boot, sometimes it takes a few tries or cycling power to get
them to boot.
___end___
Thu Mar 17 18:03:29 GMT 2011    CHIP     End LSD
Thu Mar 17 18:03:35 GMT 2011    CHIP     BiasLSD
WEATHER COMMENT: Thu Mar 17 18:03:00 GMT 2011
Still in cirrus, can’t run comp and mk4.
___end___
Thu Mar 17 18:04:19 GMT 2011    CHIP     End BiasLSD
Thu Mar 17 18:04:26 GMT 2011    CHIP     Bias
Thu Mar 17 18:05:08 GMT 2011    CHIP     End Bias
Thu Mar 17 18:05:14 GMT 2011    CHIP     ReStart Patrol
Thu Mar 17 19:28:41 GMT 2011: PSPT Start Patrol
NOTE BY ALLEN: Thu Mar 17 20:57:33 GMT 2011
Reconfigured dome shutter.
___end___
Thu Mar 17 21:37:33 GMT 2011: PSPT Start Patrol
**PSPT PROBLEM COMMENT BY ALLEN**: Thu Mar 17 21:38:00 GMT 2011
Crashed about 2130UT, killed and restarted.
___end___
Thu Mar 17 23:02:40 GMT 2011    CHIP     LSD
Thu Mar 17 23:04:12 GMT 2011    CHIP     End LSD
Thu Mar 17 23:04:18 GMT 2011    CHIP     BiasLSD
Thu Mar 17 23:05:04 GMT 2011    CHIP     End BiasLSD
Thu Mar 17 23:05:12 GMT 2011    CHIP     Bias
Thu Mar 17 23:05:40 GMT 2011    CHIP     End Bias
Thu Mar 17 23:05:45 GMT 2011    CHIP     ReStart Patrol
Fri Mar 18 00:20:08 GMT 2011    MKIV    Start Patrol
Fri Mar 18 00:51:36 GMT 2011    MKIV    End Patrol
Fri Mar 18 01:52:34 GMT 2011Fri Mar 18 01:52:26 GMT 2011        CHIP     End Patrol
: PSPT Abort Patrol



Fri Mar 18 01:54:28 GMT 2011 H-ALPHA End Patrol
WEATHER COMMENT: Fri Mar 18 01:55:34 GMT 2011
In denser cloud band, instruments idled.
___end___
CoMP COMMENT BY ALLEN: Fri Mar 18 02:05:16 GMT 2011
Taking HDD to fedex today.
___end___
Fri Mar 18 02:06:00 GMT 2011 
        MkIV    
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